

Manuscript EMBOR-2012-35709

**Nuclear movement during myotube formation is microtubule and dynein dependent and is regulated by Cdc42, Par6 and Par3**

Bruno Cadot, Vincent Gache, Elena Vasyutina, Sestina Falcone, Carmen Birchmeier and Edgar R. Gomes

*Corresponding author: Edgar R. Gomes, UMR S 787 INSERM, Universite Pierre et Marie Curie Paris 6*

---

**Review timeline:**

Submission date:

04 January 2012

Accepted:

21 May 2012

---

**Transaction Report:**

No Peer Review Process File is available with this article, as the authors have chosen not to make the review process public in this case.